Author's response to reviews

Title: MicroRNA-21 Is a Candidate Driver Gene for 17q23-25 Amplification in Ovarian Clear Cell Carcinoma

Authors:

Yukihiro Hirata (deepblue8008@gmail.com)
Noriyuki Murai (nmurai@jikei.ac.jp)
Nozomu Yanaihara (yanazou@jikei.ac.jp)
Misato Saito (misato@jikei.ac.jp)
Motoaki Saito (motoaki@jikai.ac.jp)
Mitsuyoshi Urashima (urashima@jikei.ac.jp)
Yasuko Murakami (yasukomu@jikei.ac.jp)
Senya Matsufuji (senya@jikei.ac.jp)
Aikou Okamoto (aiko@jikei.ac.jp)

Version: 2 Date: 29 September 2014

Author's response to reviews: see over
September 29, 2014

Editor-in-Chief

*BMC cancer*

We are re-submitting our manuscript entitled “*MicroRNA-21 Is a candidate Driver Gene for 17q23-25 Amplification in Ovarian Clear Cell Carcinoma*” (MS: 1393691886117873) to *BMC Cancer* as a research article.

We have made modifications to address the editorial requests. We also have corrected typographical errors in our manuscript.

We hope the manuscript now meets the criteria for publication in *BMC Cancer*. Thank you for your expert editorial guidance.

Sincerely,

Nozomu Yanaihara M.D., Ph.D.

Department of Obstetrics and Gynecology,
The Jikei University School of Medicine,
3-25-8, Nishi-Shinbashi, Minato-ku, Tokyo
105-8461, Japan

Tel: +81 3-3433-1111 (Ext. 3520)

Fax: +81 3-3433-1219

E-mail: yanazou@jikei.ac.jp